Author's response to reviews

Title: Study protocol for a cluster randomized controlled trial comparing group MEmory Specificity Training (MEST) to group education and support (ES) in the treatment of recurrent depression

Authors:

Tim Dalgleish (tim.dalgleish@mrc-cbu.cam.ac.uk)
Anna Bevan (anna.bevan@mrc-cbu.cam.ac.uk)
Anna McKinnon (anna.mckinnon@mrc-cbu.cam.ac.uk)
Lauren Breakwell (lauren.breakwell@mrc-cbu.cam.ac.uk)
Viola Mueller (viola.mueller@mrc-cbu.cam.ac.uk)
Isobel Chadwick (isobel.chadwick@mrc-cbu.cam.ac.uk)
Susanne Schweizer (susanne.schweizer@mrc-cbu.cam.ac.uk)
Peter Watson (peter.watson@mrc-cbu.cam.ac.uk)
Filip Raes (filip.raes@ppw.kuleuven.be)
Laura Jobson (laura.jobson@mrc-cbu.cam.ac.uk)
Aliza Werner-Seidler (aliza.werner-seidler@mrc-cbu.cam.ac.uk)

Version: 2
Date: 1 July 2014

Author's response to reviews:

Dr Altman, Dr Furberg and Dr Grimshaw
Editors-in-Chief
Trials
1st July 2014

Dear Editors,

Please find attached a manuscript entitled ‘A comparison of MEmory Specificity Training (MEST) to education and support (ES) in the treatment of recurrent depression: study protocol for a cluster randomized controlled trial’

We have addressed each of the minor formatting requests, with the exception of an aspect of Point 7. Specifically, as our consort diagram does not require a legend, this request did not appear applicable to our submission. Please let us know if there are further changes you require.

We would appreciate your consideration of this protocol for publication in BMC Trials.

We verify that this protocol has not been published previously.

Best wishes,
Tim Dalgleish and Aliza Werner-Seidler